Author's response to reviews

Title: Effects of Stemmed and Nonstemmed Hip Replacement on Stress Distribution of Proximal Femur and Implant

Authors:

Chun-Ming Chen (juneming1012@gmail.com)
Wen-Chi Tsai (Dtorth05@yahoo.com.tw)
Shang-Chih Lin (orthodent.cax@gmail.com)
Ching-Shiow Tseng (cstseng@cc.ncu.edu.tw)

Version: 5 Date: 28 August 2014

Author's response to reviews: see over
Author's response to reviews

Title: Effects of Stemmed and Nonstemmed Hip Replacement on Stress Distribution of Proximal Femur and Implant

Authors:

   Chun-Ming Chen (juneming1012@gmail.com)
   Wen-Chi Tsai (dtorth05@yahoo.com.tw)
   Shang-Chih Lin (orthodent.cax@gmail.com)
   Ching-Shiow Tseng (cstseng@cc.ncu.edu.tw)

Version: 5 Date: 28 August 2014

Author's response to reviews: see over
Editorial Comments:

In the methods section, can you please provide the full name and affiliation of the ethical committee that confirmed that this study did not require ethical approval. Can you please also include information regarding any legislation or laws that confirm such studies are exempt from formal ethical approval in Taiwan.

1 Thank for the editorial’s comments. The information of the ethical committee and the legislation have been given in the Methods section in the revised content (page 6, lines 14-16).